Computer Science 34

Collaboration Policy Spring 2005

IMPORTANT: Please read this document and sign below. Return it to a TA after class. You will not receive credit for any assignments until we have this on file.

CS 34 has no assignments outside of lab and is S/NC. Therefore, the collaboration policy for this course has to do with the level of in-lab collaboration allowed. Here are the basic principles on collaboration: You must:

• do your own thinking, your own design, and your own coding • be able to explain every line of code you write You may:

• talk to other students about concepts in lectures and labs • answer questions as long as the answer does not help in the problem-solving process at large You may not:

• copy another student’s code • debug another student’s code during or in between labs • discuss an assignment with someone else before you have analyzed the problem in depth on your own

If you do not know if actions are acceptable under this policy, please ask a TA. We will be happy to clarify any of the above points or any you do not think are covered.

I have read and agree to abide by the above policy

Name (please print) Signature

Date CS Login